Author's response to reviews

Title: A Cervical Ligamentum Flavum Cyst in an 82-yeal-old female presenting with spinal cord compression: Case Report and Literature Review

Authors:

Alexandros G. Brotis MD, PhD (alexgbrodis@yahoo.com)
Eftychia Z. Kapsalaki MD, PhD (ekapsal@med.uth.gr)
Evangelos K. Papadopoulos MD (neuroevan@yahoo.gr)
Kostas N. Fountas MD, PhD (fountas@med.uth.gr)

Version: 3 Date: 17 January 2012

Author's response to reviews: see over
Dear Editor,

We read your comments with great interest. All specific issues raised were addressed.

More specifically:

1. The patient’s ethnicity has been added in the “case report” section”. She is an 82-year old Greek female who presented with spinal cord compression.

2. The new running title has been modified according to the journal’s format: 'A presenting in B with C: a case report. Its revised form is as follows:

“A Cervical Ligamentum Flavum Cyst in an 82-yeal-old female presenting with spinal cord compression: Case Report and Literature Review”

Thanks in advance for your kind attention. Awaiting for your response.

Sincerely,

Alexandros G. Brotis